Author’s response to reviews

Title: Risk Factors Associated with Poor Health Outcomes for Children under the Age of 5 with Moderate Acute Malnutrition in Rural Fagita Lekoma District, Awi Zone, Amhara, Ethiopia, 2016

Authors:
Dube Jara (jaradube@yahoo.com)
W/Amilak Adamu (woldeamlakadamu@gmail.com)
Mulunesh Alemayehu (mulu2019@gmail.com)
Sahai Burrowes (sahai.burrowes@tu.edu)

Version: 4 Date: 27 Nov 2017

Author’s response to reviews:

Thank you very much for your all constructive feedback.

For the points suggested by reviewer 2

1. Page 12, Line 15: "20 higher" would be incorrect. Since AHR=2.19, the risk of poor health outcomes among children born of pregnancies that were both unwanted and unplanned was "2.19 times" (or "119% higher than") the risk among children born of wanted and/or planned pregnancies. Figure 1: Consider replacing the label "Till when did she exclusively breastfeed?" by a more standard "Exclusive breastfeeding duration".

- They were well addressed

Concerning language edition by native speaker-we did it well. As one of the author is native speaker from USA California we believe that we can do it by ourselves. Therefore, we made a few important edition from the previous.

Thank you again